\documentclass{mc}
%%%%% journal  info -- DO NOT CHANGE %%%%%%%%%
\setcounter{page}{83}
\renewcommand\thisnumber{x}
\renewcommand\thisyear {2019}
\renewcommand\thismonth{xxx}
\renewcommand\thisvolume{24}
\renewcommand\datereceived{May 17, 2018}
\renewcommand\dateaccepted{July 24, 2018}
%%%%%%    end   %%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%

\newcommand{\�}{\`a}
\newcommand{\�}{\'a}
\newcommand{\�}{\"a}
\newcommand{\�}{\`e}
\newcommand{\�}{\'e}
\newcommand{\�}{\"e}
\newcommand{\�}{\`\i}
\newcommand{\�}{\'\i}
\newcommand{\�}{\`o}
\newcommand{\�}{\'o}
\newcommand{\�}{\"o}
\newcommand{\�}{\`u}
\newcommand{\�}{\'u}
\newcommand{\�}{\"u}
\newcommand{\�}{\`E}

\newcommand{\qqed}{\tag*{\qed}} %%%
\newcommand{\qdp}{\hspace*{1.5mm}}%
\newcommand{\qqdp}{\hspace*{2.5mm}}
\newcommand{\xqdp}{\hspace*{5.0mm}}
\newcommand{\xxqdp}{\hspace*{10mm}}
\newcommand{\xquad}{\quad\qquad\quad}
\newcommand{\qdn}{\hspace*{-1.5mm}}
\newcommand{\qqdn}{\hspace*{-2.5mm}}
\newcommand{\xqdn}{\hspace*{-5.0mm}}
\newcommand{\xxqdn}{\hspace*{-10mm}}

\newcommand{\fns}{\footnotesize}
\newcommand{\dst}{\displaystyle}
\newcommand{\sst}{\scriptstyle}
\newcommand{\sss}{\scriptscriptstyle}


\newcommand{\+}{&\qqdn}%
\newcommand{\zero}{&\qdn&\xqdn}
\newcommand{\tagg}[2]{\addtocounter{equation}{#1}
		\tag{\theequation#2}}%%%\tag{}\tag*{}


\newcommand{\la}{\leftarrow}%
\newcommand{\ra}{\rightarrow}
\newcommand{\La}{\Leftarrow}%
\newcommand{\Ra}{\Rightarrow}
\newcommand{\lla}{\longleftarrow}%
\newcommand{\lra}{\longrightarrow}
\newcommand{\Lla}{\Longleftarrow}%
\newcommand{\Lra}{\Longrightarrow}
\newcommand{\mto}{\mapsto}
\newcommand{\rmap}{\longmapsto}%%%
\newcommand{\ltor}{\rightleftharpoons}

\newcommand{\sof}[1]{\lceil{#1}\rceil}
\newcommand{\pav}[1]{\lfloor{#1}\rfloor}
\newcommand{\soff}[1]{\left\lceil{#1}\right\rceil}
\newcommand{\pavv}[1]{\left\lfloor{#1}\right\rfloor}
%\newcommand{\dvv}[2]{#1\not\:|\:#2}
\newcommand{\cng}[3]{#1\not\equiv_{#2}#3}
\newcommand{\xcng}[3]{\chi(#1\not\equiv_{#2}#3)}

%\newcommand{\vec}[1]{\bar{#1}}=\vec{#1}
\newcommand{\diam}{\diamond}
\newcommand{\ang}[1]{\langle{#1}\rangle}
\newcommand{\res}[2]{\mbox{Res}\left[#1\right]_{#2}}

\newcommand{\mb}[1]{\mathbb{#1}}
\newcommand{\mc}[1]{\mathcal{#1}}
\newcommand{\mf}{\mathbf}
\newcommand{\mk}{\mathfrak}
\newcommand{\sbs}[1]{_{\substack{#1}}}
\newcommand{\sps}[1]{^{\substack{#1}}}
%\newcommand{\sopra}[2]{\overset{#1}{#2}}
%\newcommand{\sotto}[2]{\underset{#1}{#2}}
%\newcommand{\fianco}[2]{\sideset{#1}{#2}}
%\newcommad{\tilde}=wave over some thing
%\newcommand{\nabla}=backward difference operator

%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%
%%%%%%%%%%%%% divided differences %%%%%%%%%%%%%%%%%
%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%

\newcommand{\fwd}{\mbox{$\Del\qdn\!\cdot\;$}}
\newcommand{\bwd}{\nabla}
%\newcommand{\dd}[1]{\raisebox{-1mm}{\text{\huge
%%                  $\triangle$}\!}\qqdn{\sss#1}\:}
\newcommand{\dd}[2]{\mult{c}{{\sss#2}\\[-0.3mm]
\text{\Large$\bigcirc$}\\[-4.25mm]\Delta\\[-1.3mm]{\sss#1}}\qdn}
\newcommand{\dvd}[3]
           {\sum_{i=1}^n {#1}
            \prod\sbs{j=1\\j\not=i}^n
            \frac{x_i^{\!#2}}{x_i-x_j}{#3}}
\newcommand{\mdvd}[2]
           {\sum\sbs{\Lambda\subset[n]\\
                    |\Lambda|=p}{#1}
            \prod\sbs{i\in\Lambda\\j\not\in\Lambda}
            \frac{x_i}{x_i-x_j}{#2}}

%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%
%%%%%%%%%%%% hypergeometric series %%%%%%%%%%%%%%%%
%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%

\newcommand{\fat}[3]{\left(#1;#2\right)_{#3}}
\newcommand{\ffat}[3]{\left[#1;#2\right]_{#3}}
\newcommand{\hyp}[3]{\left[\qdn\ba{#1}#2\\#3\ea\qdn\right]}
\newcommand{\hyq}[4]{\left[\qdn\ba{#1}#3\\#4\ea{\!\Big|\:#2}\right]}
\newcommand{\hyl}[4]{\left[{\qdn\ba{#1}#3\\#4\ea{#2}}\right.}
\newcommand{\hyr}[4]{\left.{\ba{#1}#3\\#4\ea}{\Big|\:#2}\right]}
\newcommand{\hyd}[6]{\left[\qdn\ba{#1}#2\\#3\ea{\!\!%
             \Big|\!\!}\ba{#4}#5\\#6\ea\qdn\right]}%%%

%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%
%%%%%%%%%%%%%% genfrac applications %%%%%%%%%%%%%%%%%%%
%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%

\newcommand{\binm}{\binom}
\newcommand{\sbnm}[2]{\Bigl(\!\ba{c}\!#1\!\\#2\ea\!\Bigr)}
\newcommand{\tbnm}[2]{\tbinom{#1}{#2}}
\newcommand{\zbnm}[2]{\genfrac{(}{)}{0mm}{2}{#1}{#2}}
%\newcommand{\xbnm}[2]{\genfrac{(}{)}{0mm}{3}{#1}{#2}}
%\newcommand{\binq}[2]{\genfrac{[}{]}{0mm}{0}{#1}{#2}}
\newcommand{\binq}[2]{{#1\brack#2}}
\newcommand{\sbnq}[2]{\Bigl[\!\ba{c}\!#1\!\\#2\ea\!\Bigr]}
\newcommand{\tbnq}[2]{\genfrac{[}{]}{0mm}{1}{#1}{#2}}
\newcommand{\zfrac}[2]{\genfrac{}{}{}{2}{#1}{#2}}
%\newcommand{\xfrac}[2]{\genfrac{}{}{}{3}{#1}{#2}}
\newcommand{\double}[2]{\genfrac{}{}{0mm}{1}{#1}{#2}}
\newcommand{\doppio}[2]{\genfrac{}{}{0mm}{2}{#1}{#2}}
\newcommand{\triple}{\qdn\ba{c}{\wr}\\[-1mm]{\wr}\\[-1mm]{\wr}\ea\qdn}
\newcommand{\quint}{\mult{c}{:\\[-2.5mm]\vdots}}

%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%
%EEEEEEEEEEEEEEEEEEEEEEEEEEEEEEEEEEEEEEEEEEEEEEEEEEEE%
%%%%%%%%%%%%%%%%%%%%% equations %%%%%%%%%%%%%%%%%%%%%%

\newcommand{\nnm}{\nonumber}
\newcommand{\be}{\begin{equation}}
\newcommand{\ee}{\end{equation}}
\newcommand{\ba}{\begin{array}}
\newcommand{\ea}{\end{array}}
\newcommand{\bmn}{\begin{eqnarray}}
\newcommand{\emn}{\end{eqnarray}}
\newcommand{\bnm}{\begin{eqnarray*}}
\newcommand{\enm}{\end{eqnarray*}}
\newcommand{\bln}{\begin{subequations}}
\newcommand{\eln}{\end{subequations}}
\newcommand{\blm}{\bln\bmn}
\newcommand{\elm}{\emn\eln}

\newcommand{\pq}[1]{\begin{equation}#1\end{equation}}
\newcommand{\pp}[1]{\begin{aligned}#1%%%%%%%%%%%%%%%%%
            \end{aligned}}  %%%aligned in equation%%%%
\newcommand{\qq}[2]{\begin{alignedat}{#1}#2%%%%%%%%%%%
            \end{alignedat}}%%%alignedat in equation%%
\newcommand{\pmq}[1]{\begin{align}#1%%%%%%%%%%%%%%%%%%
            \end{align}}     %%%align=auto-space%%%%%%
\newcommand{\pnq}[1]{\begin{align*}#1%%%%%%%%%%%%%%%%%
            \end{align*}}    %%%align=auto-space%%%%%%
\newcommand{\pmp}[2]{\begin{alignat}{#1}#2%%%%%%%%%%%%
            \end{alignat}}   %%%alignat=no-space%%%%%%
\newcommand{\pnp}[2]{\begin{alignat*}{#1}#2%%%%%%%%%%%
            \end{alignat*}}  %%%alignat=no-space%%%%%%
\newcommand{\qmp}[1]{\begin{flalign}#1%%%%%%%%%%%%%%%%
            \end{flalign}}   %%%flalign=max-space%%%%%
\newcommand{\qnp}[1]{\begin{flalign*}#1%%%%%%%%%%%%%%%
            \end{flalign*}}  %%flalign=max-space%%%%%%
\newcommand{\qsq}[1]{\begin{split}#1\end{split}}%%%%%%
\newcommand{\qxq}[1]{\begin{gathered}#1%%%%%%%%%%%%%%%
            \end{gathered}}%gathered=no& in equation%%
\newcommand{\qyq}[1]{\begin{gather}#1%%%%%%%%%%%%%%%%%
            \end{gather}}   %%%%%gather senza  & %%%%%
\newcommand{\qzq}[1]{\begin{gather*}#1%%%%%%%%%%%%%%%%
           \end{gather*}}   %%%%%gather senza & %%%%%%
\newcommand{\mult}[2]{\begin{array}{#1}#2\end{array}}%
\newcommand{\multm}[1]{\begin{multline}#1%%%%%%%%%%%%%
            \end{multline}} %%%multline senza & %%%%%%
\newcommand{\multn}[1]{\begin{multline*}#1%%%%%%%%%%%%
            \end{multline*}}%%%multline senza & %%%%%%

%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%
%FFFFFFFFFFFFFFFFFFFFFFFFFFFFFFFFFFFFFFFFFFFFFFFFFFFF%
%%%%%%%%%%%%%%%%%%%%% positions %%%%%%%%%%%%%%%%%%%%%%

\newcommand{\centro}[1]
           {\begin{center}#1\end{center}}
\newcommand{\sinistra}[1]
           {\begin{flushleft}#1\end{flushleft}}
\newcommand{\destra}[1]
           {\begin{flushright}#1\end{flushright}}
\newcommand{\centerbox}[2]{\centro{\begin{tabular}{|c|}\hline
\parbox{#1}{\mbox{}\\[2.5mm]#2\mbox{}\\}\\\hline\end{tabular}}}
\newcommand{\centrobmp}[3]{\vspace*{1mm}
		\centerbmp{#1}{#2}{#3}\vspace*{1mm}}

%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%
%GGGGGGGGGGGGGGGGGGGGGGGGGGGGGGGGGGGGGGGGGGGGGGGGGGGG%
%%%%%%%%%%%%%%%%%%% Greek letters %%%%%%%%%%%%%%%%%%%%

\newcommand{\alp}{\alpha}
\newcommand{\bet}{\beta}
\newcommand{\gam}{\gamma}

\newcommand{\lam}{\lambda}
%\newcommand{\mu}{\mu}
%\newcommand{\nu}{\nu}
%\newcommand{\rho}{\rho}
\newcommand{\vrho}{\varrho}

\newcommand{\eps}{\epsilon}
\newcommand{\veps}{\varepsilon}
\newcommand{\del}{\delta}

\newcommand{\sig}{\sigma}
%\newcommand{\tau}{\tau}

%\newcommand{\theta}{\theta}
\newcommand{\vthe}{\vartheta}
\newcommand{\ome}{\omega}

\newcommand{\ii}{\imath}
\newcommand{\jj}{\jmath}
\newcommand{\kk}{\kappa}

%\newcommand{\iota}{\iota}
%\newcommand{\kap}{\kappa}
%\newcommand{\ell}{\ell}

%\newcommand{\xi}{\xi}
%\newcommand{\zeta}{\zeta}
\newcommand{\vsig}{\varsigma}
%\newcommand{\eta}{\eta}

%\newcommand{\phi}{\phi}
\newcommand{\vph}{\varphi}
%\newcommand{\psi}{\psi}

\newcommand{\Del}{\Delta}
\newcommand{\The}{\Theta}
\newcommand{\Lam}{\Lambda}
\newcommand{\Ome}{\Omega}

%\newcommand{\Xi}{\Xi}
%\newcommand{\Pi}{\Pi}
\newcommand{\Sig}{\Sigma}

\newcommand{\Gam}{\Gamma}
%\newcommand{\Phi}{\Phi}
%\newcommand{\Psi}{\Psi}
\newcommand{\Ups}{\Upsilon}


\newtheorem{entry}{Entry}%%%%%%%%%%%%%%%%
\renewcommand{\theentry}{%\Alph{chapter}%
\arabic{section}-\arabic{entry}}%%%%%%%%%
\newcommand{\ent}[3]{\begin{entry}[#2]%%%
\label{entry:#1}\bnm#3\enm\end{entry}}%%%

%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%
%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%
%IIIIIIIIIIIIIIIIIIIIIIIIIIIIIIIIIIIIIIIIIIIIIIIIIIII%
%%%%%%%%%%%%%%% Environment Italiano %%%%%%%%%%%%%%%%%

\newcommand{\summary}[2]{\begin{center}\parbox{#1}
{{\sc\bf Summary}:\,{\small\it#2}}\end{center}}
\newcommand{\riassunto}[2]{\begin{center}\parbox{#1}
{{\sc\bf Riassunto}:\,{\small\it#2}}\end{center}}
\newcommand{\grazia}[2]{\begin{center}\parbox{#1}
{{\sc\bf Ringraziamento}:\,{\small\it#2}}\end{center}}
\newcommand{\thank}[2]{\begin{center}\parbox{#1}
{{\sc\bf Acknowledgement}:\,{\small\it#2}}\end{center}}

%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%
%%%%%MMMMMMMMMMMMMMMMMMMMMMMMMMMMMMMMMMMMMMMMMMMMMMMMMMMM%%%%%
%%%%%%%%%%%%%%%%%%%%% References %%%%%%%%%%%%%%%%%%%%%%%%%%%%%

\newcommand{\nota}[2]{\centerbox{#1}{\textbf{Achtung}\quad#2}}

\newcommand{\bbtm}[4]{\bibitem{kn:#1}{#2,}~\emph{#3,}~{#4.}}	
\newcommand{\cito}[1]{\cite{kn:#1}}	
\newcommand{\citu}[2]{\cite[#2]{kn:#1}}

\newcommand{\mr}{\textbf{MR}\:}
\newcommand{\zbl}{\textbf{Zbl}\:}

\usepackage[displaymath,mathlines,pagewise]{lineno}
\usepackage{eucal,cite}

\begin{document}
\begin{thebibliography}{99}

\bibitem{abel}{\sc N.\,H.\,Abel},
     {\em Sur les fonctions g\�n\�ratrices et leurs d\�terminants
      in his ``{\OE}uvres Compl\�tes"}, Gr{\o}ndahl \& Son
     Forlag, Oslo, 1881.

\bibitem{chu10f}{\sc X.\,Chen, W.\,Chu},
     {\em Dixon's $_3F_2(1)$-series and identities involving
      harmonic numbers and the Riemann zeta function},
     Discrete Math. {\bf 310}(2010), 83--91.

\bibitem{chu98b}{\sc W.\,Chu},
     {\em Generating functions and combinatorial identities},
     Glas. Mat. Ser. III {\bf 33}(1998), 1--12.

\bibitem{chu99a}{\sc W.\,Chu},
     {Finite differences and orthogonal polynomials},
     The Ramanujan J. {\bf 3}(1999).

\bibitem{chu02a}{\sc W.\,Chu},
     {\em Some binomial convolution formulas},
     Fibonacci Quart. {\bf 40}(2002), 19--32.

\bibitem{chu06f}{\sc W.\,Chu},
     {\em The Fa\`a Di Bruno formula and determinant identities},
     Linear Multilinear Algebra {\bf 54}(2006), 1--25.

\bibitem{chu10b}{\sc W.\,Chu},
     {\em Elementary proofs for convolution identities of Abel and Hagen--Rothe},
     Electron. J. Combin. {\bf 17}(2010), \#N24.

\bibitem{chu13a}{\sc W.\,Chu},
     {\em Derivative inverse series relations and Lagrange expansion formula},
     Int. J. Number Theory {\bf 9}(2013), 1001--1013.

\bibitem{comtet} {\sc L. Comtet},
     {\em Advanced combinatorics}, Dordrecht--Holland, Dodrecht, 1974.

\bibitem{gould}{\sc H. W. Gould},
     {\em Some generalizations of Vandermonde's convolution}, Amer. Math. Monthly {\bf 63}(1956), 84--91.

\bibitem{graham}{\sc R.\,L.\,Graham, D.\,E.\,Knuth, O.\,Patashnik},
     {\em Concrete mathematics},
     Addison-Wesley Publ.\,Company, Reading, Massachusetts, 1989.

\bibitem{koshy}{\sc T. Koshy},
     {\em Catalan numbers with applications},
     Oxford University Press, New York, 2009.

\bibitem{knuth}{\sc D.\,E.\,Knuth},
     {\em Log--squared of the Catalan generating function},
     Amer. Math. Monthly {\bf 122}(2015), 390.
     %% (Solution) 124:7(2017), 660--661}

\bibitem{lag}{\sc J.\,L.\,Lagrange},
     {\em Nouvelle m\�thode pour r\�soudre les \�quations litt\�rales par le moyen des s\�ries}
      {M\�moires de l'Acad\�mie Royale des Sciences et Belles--Lettres de Berlin} {\bf 24}(1770).

\bibitem{lam-a}{\sc J.\,H.\,Lambert},
     {\em Observationes variae in Mathesin puram},
     Acta Helvetica {\bf 3}(1758), 128--168;
      reprinted in Opera Mathematica, 16--51.

\bibitem{lam-b}{\sc J.\,H.\,Lambert},
     {\em Observationes analytiques},
     Nouveaux M\�moires de l'Acad\�mie royale des Sciences
      et Belles--Lettres Berlin (1770), 225--244;
      reprinted in Opera Mathematica, 270--290.

\bibitem{mac}{\sc I.\,G.\,Macdonald},
     {\em Symmetric functions and Hall polynomials},
     Clarendon, Oxford, 1979.

\bibitem{riordan} {\sc J.\,Riordan},
     {\em Combinatorial identities},
     John Wiley \& Sons, New York, 1968.

\bibitem{stanley}{\sc R.\,P.\,Stanley},
     {\em Enumerative combinatorics},
     Wadsworth and Brooks/Cole, Monterey, CA, 1999.

\bibitem{stanley-catalan}{\sc R.\,P.\,Stanley},
     {\em Catalan numbers},
     Cambridge University Press, New York, 2015.
\end{thebibliography}

%% \begin{linenumbers}

\markboth{W.\,Chu}{Logarithms of a Binomial series}

\title{Logarithms of a binomial series: Extension of a series of Knuth}

\author[W.\,Chu]{Wenchang Chu\corrauth}

\address{School of Mathematics and Statistics, Zhoukou Normal University, 466\,001 Henan, P.\,R.\,China}

\email{{\tt chu.wenchang@unisalento.it} (W.\,Chu)}

\begin{abstract}
The Lagrange expansion formula is employed to determine
the Maclaurin series for the logarithms of Lambert series
of binomial coefficients, extending the log--squared
of the Catalan generating function due to Knuth (2015).
\end{abstract}

\keywords{Catalan numbers, harmonic numbers, elementary symmetric function, Bell polynomial, Lagrange expansion formula}

\ams{Primary 05A15; Secondary 05A10}

\maketitle

\section{Introduction and motivation}
%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%

Let $\mb{N}$ be the set of natural numbers with
$\mb{N}_0=\{0\}\cup\mb{N}$. Catalan numbers
(cf.~\cite{koshy}) defined by
\pq{C_n=\frac1{n+1}\binm{2n}{n}
\quad\text{with}\quad n\in\mb{N}_0}
are one of the fascinating sequences
in classical combinatorics with more
than fifty significant combinatorial
interpretations~\cite{stanley,stanley-catalan}.
Their ordinary generating function is given
by
\[
C(x)=\sum_{n\ge0}C_nx^n=\frac{1-\sqrt{1-4x}}{2x}.
\]
Recently, Knuth~\cite{knuth} discovered the following
remarkable log--squared series
\pq{\log^2C(x)=\label{gf-2c}
\sum_{m\ge2}\frac{x^m}{m}
\binm{2m}{m}\big\{H_{2m-1}-H_m\big\}, }
where   harmonic numbers are defined by
\[
H_m:=H_m^{\sss\ang{1}}
\qquad\text{with}\qquad
H_m^{\sss\ang{n}}
=\sum_{k=1}^m\frac1{k^n}.
\]
The aim of this short paper is to extend Knuth's identity
to Lambert's binomial series (see~\cite{lam-a,lam-b})
with higher powers. The main theorem will be proved in the
next section by means of the Lagrange expansion formula.
Then the paper will end with a few applications to convolution
identities involving harmonic numbers.

In order to facilitate the subsequent references,
we record briefly the elementary symmetric functions
and their representation in terms of power sum functions.
For an indeterminate $y$ and $m,n\in\mb{N}$,
define the power sums
\[
\mc{P}_n(y|m):=\sum_{i=1}^m\frac{1}{(y+i)^{n}}
\]
and the elementary symmetric functions
\[
\Ome_{0}(y|m)=1
\quad\text{and}\quad
\Ome_{n}(y|m)
:=\qqdn\sum_{1\le k_1<k_2<\cdots<k_n\le m}
\prod_{i=1}^{n}\frac1{y+k_i}.
\]
Their generating function can be manipulated as follows:
\pnq{\sum_{n=0}^{m}x^n\Ome_{n}(y|m)
=&\prod_{k=1}^m\Big(1+\frac{x}{y+k}\Big)\\
=&\exp\bigg\{\sum_{k=1}^m\log\Big(1+\frac{x}{y+k}\Big)\bigg\}\\
=&\exp\bigg\{\sum_{k=1}^m\sum_{i=1}^{\infty}
\frac{(-1)^{i-1}x^i}{i(y+k)^{i}}\bigg\}\\
=&\exp\bigg\{\sum_{i=1}^{\infty}x^i
\frac{(-1)^{i-1}\mc{P}_i(y|m)}{i}\bigg\}.}
By extracting the coefficient of $x^n$ from the two extreme
members of these equations, we obtain the expression
in terms of power sums (cf. \cite[\S1.2]{mac}
and \cite{chu99a,chu06f,chu10f})
\pq{\Ome_{n}(y|m)
=\sum_{\sig(n)}
(-1)^n\prod_{i=1}^n
\frac{\mc{P}_i^{j_i}(y|m)}{j_i!}
\Big(\frac{-1}{i}\Big)^{j_i},}
where the multiple sum runs over all the partitions
$\sig(n)$ of $n$ indexed by nonnegative integers
$\{j_1,j_2,\cdots,j_n\}$ subject to the condition
$j_1+2j_2+\cdots+nj_n=n$. Denote further by $\sig_{k}(n)$
the subset of partitions of $n$ with exactly $k$
parts, characterized by $\{j_1,j_2,\cdots,j_n\}$
subject to $j_1+2j_2+\cdots+nj_n=n$ and
$j_1+j_2+\cdots+j_n=k$. Then the exponential
Bell polynomials (see \cite[\S3.3]{comtet})
read:
\[\mf{Y}_n(x_1,x_2,\cdots,x_n)
:=\sum_{k=0}^n
\mf{B}_{n,k}(x_1,x_2,\cdots,x_{n-k+1})\]
with the partial Bell polynomials   given
by $\mf{B}_{0,0}(x_1)=1$ and
\pnq{\mf{B}_{n,k}(x_1,x_2,\cdots,x_{n-k+1})
:=&\frac{n!}{k!}\sum_{\sig_{k}(n)}
\binm{k}{j_1,j_2,\cdots,j_n}
\prod_{i=1}^n\Big(\frac{x_i}{i!}\Big)^{j_i}\\
=&\frac{1}{k!}\sum\sbs{i_1+i_2+\cdots+i_k=n\\(i_j\in\mb{N})}
\binm{n}{i_1,i_2,\cdots,i_k}
\prod_{j=1}^kx_{i_j}.}

We have therefore the following alternative expression
of $\Ome_{n}(y|m)$ by the complete Bell polynomials:
\pq{\pp{&\Ome_{n}(y|m)
=\frac1{n!}\times
\mf{Y}_n(\mc{Q}_1,\mc{Q}_2,\cdots,\mc{Q}_n),\\
\text{where}\qquad
&\mc{Q}_k:=\mc{Q}_k(y|m)
=(-1)^{k-1}(k-1)!\mc{P}_k(y|m).}}

It may also be worth remarking that the last formula
for $\Ome_{n}$ is effectively a sum over the permutations
of $\{1,2,\cdots,n\}$. For example, there are 6 permutations
of $\{1,2,3,4\}$ consisting of a single cycle, 1 permutation
with four cycles of length 1, and $\binm42=6$ permutations
with cycle lengths (2,1,1). There are two different kinds
of permutations of $\{1,2,3,4\}$ with two cycles; 3 with
cycle lengths (2,2) and $2\binm43=8$ permutations with
cycle lengths (3,1). If we attach $(-1)^k$ to each
of these coefficients, where $k$ is the number of cycles,
and divide by the total number $4!=24$ of permutations,
then we get the formula for $\Ome_{4}$ in terms
of $\mc{P}_1,\mc{P}_2,\mc{P}_3,\mc{P}_4$.

The first five expressions are displayed below
with the common parameters $(y|m)$ being suppressed:
\[\pp{
\Ome_1=&\mc{P}_1,\\
\Ome_2=&\frac{1}{2}(\mc{P}_1^2-\mc{P}_2),\\
\Ome_3=&\frac{1}{6}(\mc{P}_1^3-3\mc{P}_1\mc{P}_2+2\mc{P}_3),\\
\Ome_4=&\frac{1}{24}(\mc{P}_1^4-6\mc{P}_1^2\mc{P}_2
	+8\mc{P}_1\mc{P}_3+3\mc{P}_2^2-6\mc{P}_4),\\
\Ome_5=&\frac{1}{120}
        (\mc{P}_1^5-10\mc{P}_1^3\mc{P}_2+20\mc{P}_1^2\mc{P}_3
	+15\mc{P}_1\mc{P}_2^2-30\mc{P}_1\mc{P}_4-20\mc{P}_2\mc{P}_3+24\mc{P}_5).}\]


\section{Main theorem and proof}
%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%
%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%
In classical analysis and enumerative combinatorics,
the Lagrange expansion formula is fundamental.
%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%
\begin{lemma}[\text{Lagrange expansion formula \cite{lag}: see \cite[\S3.8]{comtet} and \cite{chu98b,chu13a}}]
For a formal power series  $\vph(x)$ subject to $\vph(0)\not=0$,
the functional equation $x=y/\vph(y)$ determines $y$
as an implicit function of $x$. Then for another formal
power series $F(y)$ in the variable $y$, the following
expansions hold for both composite series:
\pmq{
F(y(x))&=\label{lag-inv}F(0)+\sum_{n=1}^{\infty}
\frac{x^n}{n}[y^{n-1}]\big\{F'(y)\vph^n(y)\big\},\\
\frac{F(y(x))}{1-\big(y\vph'(y)/\vph(y)\big)}
&=\label{lag+inv}
\sum_{n=0}^{\infty}{x^n}
[y^{n}]\big\{F(y)\vph^n(y)\big\}.}
\end{lemma}

By taking $F(y)=(1+y)^{\alp}$ and $\vph(y)=(1+y)^{\lam}$
in this lemma, we recover the binomial series
due to Lambert \cite{lam-a,lam-b}
(see also \cite{chu98b,chu02a,gould}, \cite[\S5.4]{graham}
and \cite[\S5.4]{riordan}):
\[\left.\pp{
&\sum_{k\ge0}\frac{\alp}{\alp+k\lam}
\binm{\alp+k\lam}{k}x^k
=(1+y)^{\alp}\\
&\sum_{k\ge0}
\binm{\alp+k\lam}{k}x^k
=\frac{(1+y)^{\alp+1}}{1+y-\lam y}}
\right\}\quad\text{where}\quad
y:=y_{\lam}(x)=x(1+y)^{\lam}.\]
Taking $\alp=1$ now gives the generating function
\[C_{\lam}(x):=1+y_\lam(x)
=\sum_{n\ge0}\binm{1+n\lam}{n}
\frac{x^n}{1+n\lam}\]
of the extended Catalan numbers, where $y=x(1+y)^{\lam}$.
In particular, we have the generating function
$C(x)=1+y_2(x)$ for the usual Catalan numbers.

By making use of \eqref{lag-inv}, we can expand,
under the specifications $F(y)=\log(1+y)$ and
$\vph(y)$ as before, the logarithm function
\pq{\log C_{\lam}(x)=\label{log-1}
\sum_{m\ge1}\frac{x^m}{m\lam}\binm{m\lam}{m}.}
Furthermore, we can show the following main theorem.
%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%
\begin{theorem} \label{log-n} Let $n \in\mb{N}$ and $\lam\in\mb{N}$. It holds:
\[
\log^nC_\lam(x)=n!\sum_{m\ge n}
\frac{x^m}{m\lam}\binm{m\lam}{m}
\Ome_{n-1}(m\lam-m|{m-1}).
\]
\end{theorem}

\proof
The formula in the theorem is clearly true for $n=1$
in view of \eqref{log-1}. Suppose that the formula
is valid for $n$. Then we have to prove it for $n+1$.
Because
\[
\log^nC_\lam(x)\times\log C_\lam(x)=\log^{n+1}C_\lam(x),
\]
the formula to be shown is equivalent to %% the following one
\pq{\sum_{k=n}^{m-1}
\frac{\Ome_{n-1}(k\lam-k|{k-1})}{k\lam}\binm{k\lam}{k}
\frac{\binm{m\lam-k\lam}{m-k}}{m\lam-k\lam}
=\label{check}
\frac{n+1}{m\lam}\binm{m\lam}{m}
\Ome_{n}(m\lam-m|{m-1}).}

Recall the Hagen--Rothe convolution formula
(cf. \cite{chu10b} and \cite{gould})
\[\sum_{k=0}^m\frac{x}{x+k\lam}\binm{x+k\lam}{k}
\frac{y}{y+m\lam-k\lam}\binm{y+m\lam-k\lam}{m-k}
=\frac{x+y}{x+y+m\lam}\binm{x+y+m\lam}{m}.\]
Moving the $k=m$ term to the other side, dividing by $y$,
and letting $y\to0$, we get
\pnq{\sum_{k=0}^{m-1}
\frac{x\binm{x+k\lam}{k}}{x+k\lam}
\frac{\binm{m\lam-k\lam}{m-k}}{m\lam-k\lam}
=&\lim_{y\to0}
\Bigg\{\frac{(x+y)\binm{x+y+m\lam}{m}}{y(x+y+m\lam)}
-\frac{x\binm{x+m\lam}{m}}{y(x+m\lam)}\Bigg\} }
%%[\\frac{\binm{x+m\lam}{m}}{x+m\lam}
%%\bigg\{1+x\Ome_1(x+m\lam-m|{m-1})\bigg\}\]
which can be reformulated equivalently as the equality
\pq{\sum_{k=0}^{m-1}
\frac{x\binm{x+k\lam}{k}}{x+k\lam}
\frac{\binm{m\lam-k\lam}{m-k}}{m\lam-k\lam}
=\label{b-rr}
\bigg\{1+x\frac{\text{d}}{\text{d}x}\bigg\}
\frac{\binm{x+m\lam}{m}}{x+m\lam}.}
Recalling the generating function
\[\sum_{n=0}^{m-1}x^n\Ome_{n}(y|{m-1})
=\prod_{k=1}^{m-1}\Big(1+\frac{x}{y+k}\Big)
=\frac{y+m}{x+y+m}\frac{\binm{x+y+m}{m}}{\binm{y+m}{m}}\]
we have
\[[x^n]\frac{\binm{x+y+m}{m}}{x+y+m}
=\frac{\binm{y+m}{m}}{y+m}\Ome_{n}(y|{m-1}),\]
where $[x^n]\psi(x)$ stands for the coefficient
of $x^n$ in the formal power series $\psi(x)$.
Then the coefficient of $x^n$ from the sum
on the left-hand side of \eqref{b-rr} reads:
\pnq{[x^n]\sum_{k=0}^{m-1}
\frac{x\binm{x+k\lam}{k}}{x+k\lam}
\frac{\binm{m\lam-k\lam}{m-k}}{m\lam-k\lam}
=&\sum_{k=0}^{m-1}
\frac{\binm{m\lam-k\lam}{m-k}}{m\lam-k\lam}
[x^{n-1}]\frac{\binm{x+k\lam}{k}}{x+k\lam}\\
=&\sum_{k=0}^{m-1}
\frac{\Ome_{n-1}(k\lam-k|{k-1})}{k\lam}\binm{k\lam}{k}
\frac{\binm{m\lam-k\lam}{m-k}}{m\lam-k\lam}.}
The coefficient of $x^n$ from the right-hand side member
of \eqref{b-rr} can be extracted similarly��
\pnq{[x^n]\bigg\{1+x\frac{\text{d}}{\text{d}x}\bigg\}
\frac{\binm{x+m\lam}{m}}{x+m\lam}
=&[x^n]\frac{\binm{x+m\lam}{m}}{x+m\lam}
+n[x^{n}]\frac{\binm{x+m\lam}{m}}{x+m\lam}\\
=&\frac{n+1}{m\lam}\binm{m\lam}{m}\Ome_{n}(m\lam-m|{m-1}).}
Therefore \eqref{check} is verified and Theorem~\ref{log-n}
follows by induction.
\endproof


\section{Examples as applications}
%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%
%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%


By making use of the relation
\[
\mc{P}_n(y|m)
=H_{m+y}^{\sss\ang{n}}
-H_{y}^{\sss\ang{n}}
\quad\text{for}\quad
y\in\mb{N}_0,
\]
from Theorem~\ref{log-n} we can derive
the following infinite series expressions.
%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%
\begin{proposition} \label{log<5} Let $\lam\in\mb{N}$. It holds:
\pnq{
\log^2C_\lam(x)=2\sum_{m\ge2}
\frac{x^m}{m\lam}\binm{m\lam}{m}
&\big\{H_{m\lam-1}-H_{m\lam-m}\big\},\\
\log^3C_\lam(x)=3\sum_{m\ge3}
\frac{x^m}{m\lam}\binm{m\lam}{m}
&\Big\{\big(H_{m\lam-1}-H_{m\lam-m}\big)^2
-\big(H^{\sss\ang{2}}_{m\lam-1}
-H^{\sss\ang{2}}_{m\lam-m}\big)\Big\},\\
\log^4C_\lam(x)=4\sum_{m\ge4}
\frac{x^m}{m\lam}\binm{m\lam}{m}
&\Bigg\{\mult{r}{
(H_{m\lam-1}-H_{m\lam-m}\big)^3
+2\big(H^{\sss\ang{3}}_{m\lam-1}
-H^{\sss\ang{3}}_{m\lam-m}\big)\\[2mm]
-3\big(H_{m\lam-1}-H_{m\lam-m}\big)
\big(H^{\sss\ang{2}}_{m\lam-1}
-H^{\sss\ang{2}}_{m\lam-m}\big)}
\Bigg\}.}
\end{proposition}

In particular, we recover, for $\lam=2$, Knuth's series
\eqref{gf-2c} and two further  series of cubic and quartic
logarithms of the Catalan generating function in terms
of generalized harmonic numbers.
%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%
\begin{corollary}\label{log>5}
\pnq{
\log^3C(x)=3\sum_{m\ge3}
\frac{x^m}{2m}\binm{2m}{m}
&\Big\{\big(H_{2m-1}-H_{m}\big)^2
-\big(H^{\sss\ang{2}}_{2m-1}
-H^{\sss\ang{2}}_{m}\big)\Big\},\\
\log^4C(x)=4\sum_{m\ge4}
\frac{x^m}{2m}\binm{2m}{m}
&\Bigg\{\mult{r}{
(H_{2m-1}-H_{m}\big)^3
+2\big(H^{\sss\ang{3}}_{2m-1}
-H^{\sss\ang{3}}_{m}\big)\\[2mm]
-3\big(H_{2m-1}-H_{m}\big)
\big(H^{\sss\ang{2}}_{2m-1}
-H^{\sss\ang{2}}_{m}\big)}
\Bigg\}.}
\end{corollary}

By applying the exponential law to Theorem~\ref{log-n},
we get the next convolution formula.
%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%
\begin{proposition}For $\alp,\gam\in\mb{N}$ and $m\in\mb{N}$ it holds:
\pnq{\sum_{k=\alp}^{m-\gam}\binm{k\lam}{k}&
\frac{\Ome_{\alp-1}(k\lam-k|{k-1})}{k\lam}
\binm{m\lam-k\lam}{m-k}
\frac{\Ome_{\gam-1}((m-k)(\lam-1)|{m-k-1})}{(m-k)\lam}\\
=&\binm{\alp+\gam}{\alp}\binm{m\lam}{m}
\frac{\Ome_{\alp+\gam-1}(m\lam-m|{m-1})}{m\lam}.}
\end{proposition}

When $\alp=\gam=1$ and $\alp=2,\:\gam=1$, the above formula
reduces to the following two further interesting ones.
%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%
\begin{corollary}Let $\lam\in\mb{N}$ and $m\in\mb{N}$. It holds:
\pnq{\sum_{k=1}^{m-1}\binm{k\lam}{k}
&\binm{m\lam-k\lam}{m-k}
\frac{m}{k(m-k)}
=2\lam\binm{m\lam}{m}
\big\{H_{m\lam-1}-H_{m\lam-m}\big\},\\
\sum_{k=2}^{m-1}\binm{k\lam}{k}
&\binm{m\lam-k\lam}{m-k}
\frac{H_{k\lam-1}-H_{k\lam-k}}{k(m-k)}\\
=&\frac{3\lam}{2m}\binm{m\lam}{m}
\Big\{\big(H_{m\lam-1}-H_{m\lam-m}\big)^2
-\big(H^{\sss\ang{2}}_{m\lam-1}-H^{\sss\ang{2}}_{m\lam-m}\big)\Big\}.}
\end{corollary}

In addition, by combining \eqref{log-1}
with Theorem~\ref{log-n} we can derive the following multiple
convolution identity.
%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%
\begin{corollary}Let $\lam\in\mb{N}$ and $m,n\in\mb{N}$.
Denote by $h_k(\lam)$ the falling factorial
\[h_k(\lam)=(k\lam-1)(k\lam-2)\cdots(k\lam-k+1)
\quad\text{for}\quad k\ge1.\]
Then there holds the summation formula
\[h_m(\lam)\Ome_{n-1}(m\lam-m|{m-1})
=\mf{B}_{m,n}\big(h_1(\lam),h_2(\lam),\cdots,h_{m-n+1}(\lam)\big).\]
\end{corollary}

Analogously, there are the series for Abel coefficients~\cite{abel}
(see also \cite{chu98b,chu02a} for example)
\[\left.\pp{
&\sum_{k\ge0}\frac{\alp}{\alp+k\lam}
\frac{(\alp+k\lam)^k}{k!}x^k
=e^{z\alp}\\
&\sum_{k\ge0}
\frac{(\alp+k\lam)^k}{k!}x^k
=\frac{e^{z\alp}}{1-z\lam}}
\right\}\quad\text{where}\quad
z:=z_{\lam}(x)=xe^{z\lam}.\]

In this case, taking into account \eqref{lag-inv} we have the expressions
\[z_{\lam}(x)=\label{exp-1}
\sum_{m\ge1}\frac{(m\lam)^{m-1}}{m!}{x^m}
\quad\text{and}\quad
z^n_{\lam}(x)=%\label{exp-1}
\sum_{m\ge n}\frac{n}{m}
\frac{(m\lam)^{m-n}}{(m-n)!}{x^m}.\]
They lead to the following multiple convolution formula
for the number of forests of $n$ labeled rooted trees on $m$ vertices
\[\binm{m-1}{n-1}m^{m-n}
=\mf{B}_{m,n}\big(1^0,2^1,3^2,\cdots,(m-n+1)^{m-n}\big)\]
%\[\boxed{\sum\sbs{k_1+k_2+\cdots+k_n=m\\
%	k_\ii\ge1 (1\le\ii\le n)}
%\quad\prod_{\ii=1}^n
%\frac{k_{\ii}^{k_{\ii}-1}}{k_{\ii}!}
%=\frac{n}{m}\frac{m^{m-n}}{(m-n)!}}\]
as well as the convolution identity
\[
\sum_{k=\alp}^{m-\gam}\frac{\alp}{k}
\frac{k^{k-\alp}}{(k-\alp)!}
\frac{\gam}{m-k}
\frac{(m-k)^{m-k-\gam}}{(m-k-\gam)!}
=\frac{\alp+\gam}{m}
\frac{m^{m-\alp-\gam}}{(m-\alp-\gam)!},
\]
which can also be derived from the Abel identity
(see \cite{chu10b} and \cite[\S3.1]{comtet}).

\section*{Acknowledgement}
The author is grateful to an anonymous
referee for the careful reading, critical comments and
valuable suggestions that make the manuscript improved
substantially in revision.



%% \end{linenumbers}
\end{document}  